Author's response to reviews

Title: The Chernobyl Childhood Leukemia Study: Background & Lessons Learned

Authors: Martin C. Mahoney1,2, Kirsten M. Moysich1,2, Philip L. McCarthy, Jr.1, Richard C. McDonald3, Valery F. Stepanenko, PhD4, Robert W. Day, MD, PhD5, Arthur M. Michalek, PhD1,2 1Roswell Park Cancer Institute, Buffalo, NY 2School of Medicine & Biomedical Sciences, Buffalo, NY 3International Consortium for Research on the Health Effects of Radiation (ICRHER), Bartlesville, OK 4Medical Radiological Research Center, Obninsk, Russia 5Fred Hutchinson Cancer Research Center, Seattle, WA martin.mahoney@roswellpark.org kirsten.moysich@roswellpark.org philip.mccarthy@roswellpark.org richard.valeri@obninsk.com rday@fhcrc.org arthur.michalek@roswellpark.org Corresponding Author: Martin C. Mahoney, MD, PhD, Division of Cancer Prevention & Population Sciences, Roswell Park Cancer Institute, Elm and Carlton Streets, Buffalo, NY 14263; Tel: 716-845-5790; Fax: 716-845-7783; e-mail: martin.mahoney@

Authors:

Martin C. Mahoney (martin.mahoney@roswellpark.org)
Kirsten M. Moysich (kirsten.moysich@roswellpark.org)
Philip L. McCarthy (philip.mccarthy@roswellpark.org)
Richard C. McDonald (rmcdonald@icrher.com)
Valery Stepanenko (valeri@obninsk.com)
Robert W. Day (rday@fhcrc.org)
Arthur M. Michalek (arthur.michalek@roswellpark.org)

Version: 2 Date: 18 August 2004

Author's response to reviews: see over
Please change e-mail address for author #4 (Richard McDonald) to:

dickmc@bartnet.net